

# Rapid Restratification Processes Control Mixed Layer Turbulence and Phytoplankton Growth in a Deep Convection Region

Mariana Miracca-Lage, Johannes Becherer, Lucas Merckelbach, Anthony Bosse, Pierre Testor, Jeffrey Carpenter

# ▶ To cite this version:

Mariana Miracca-Lage, Johannes Becherer, Lucas Merckelbach, Anthony Bosse, Pierre Testor, et al.. Rapid Restratification Processes Control Mixed Layer Turbulence and Phytoplankton Growth in a Deep Convection Region. Geophysical Research Letters, 2024, 51 (10), 10.1029/2023gl107336. hal-04650283

# HAL Id: hal-04650283 https://hal.science/hal-04650283v1

Submitted on 17 Jul2024

**HAL** is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.



Distributed under a Creative Commons Attribution 4.0 International License

ഹി



# **Geophysical Research Letters**<sup>•</sup>

# **RESEARCH LETTER**

10.1029/2023GL107336

#### **Key Points:**

- Autonomous glider-based turbulence measurements have captured restratification in a deep convection region
- The onset of restratification occurs rapidly, over a timescale of days to one week
- Near-surface restratification is a primary control on mixed layer turbulence and phytoplankton growth

#### **Supporting Information:**

Supporting Information may be found in the online version of this article.

#### Correspondence to:

M. Miracca-Lage, mariana.lage@hereon.de

#### Citation:

Miracca-Lage, M., Becherer, J., Merckelbach, L., Bosse, A., Testor, P., & Carpenter, J. R. (2024). Rapid restratification processes control mixed layer turbulence and phytoplankton growth in a deep convection region. *Geophysical Research Letters*, *51*, e2023GL107336 https://doi.org/10.1029/2023GL107336

Received 22 NOV 2023 Accepted 4 APR 2024

#### © 2024. The Authors. This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution cad merculation in equi

distribution and reproduction in any medium, provided the original work is properly cited.

# Rapid Restratification Processes Control Mixed Layer Turbulence and Phytoplankton Growth in a Deep Convection Region

Mariana Miracca-Lage<sup>1</sup>, Johannes Becherer<sup>1,2</sup>, Lucas Merckelbach<sup>1</sup>, Anthony Bosse<sup>3</sup>, Pierre Testor<sup>4</sup>, and Jeffrey R. Carpenter<sup>1</sup>

<sup>1</sup>Institute of Coastal Ocean Dynamics, Helmholtz-Zentrum Hereon, Geesthacht, Germany, <sup>2</sup>Institute of Oceanography, Universität Hamburg, Hamburg, Germany, <sup>3</sup>Mediterranean Institute of Oceanography, CNRS, IRD, MIO, Aix Marseille University, Université de Toulon, Marseille, France, <sup>4</sup>Laboratoire d'Océanographie et de Climatologie (LOCEAN), CNRS, IRD, MNHN, UMR, Institut Pierre Simon Laplace (IPSL), Sorbonne Université (Pierre & Marie Curie Campus), Paris, France

**Abstract** The Gulf of Lion, Northwestern Mediterranean Sea, is one of few oceanic regions where deep convection occurs. We investigate the restratification following a convection event using measurements from an ocean glider equipped with turbulence microstructure sensors. This unique combination of instruments provides a high-resolution description of the mixed layer with regard to turbulence, stratification and chlorophyll. We observe a rapid restratification process that proceeds over a timescale of days to one week. We find that restratification exerts a leading order control on surface mixed layer turbulence variability, as abrupt changes in turbulence dissipation rates are associated with the formation of near-surface stratification. The near-surface formation of stratification occurs through both the diurnal variability in surface buoyancy fluxes and through lateral advective processes. We conclude that daily near-surface processes that influence stratification control mixed layer turbulence levels, and thus the phytoplankton response in the critical transition period to spring bloom.

**Plain Language Summary** During winter in the Gulf of Lion, Northwestern Mediterranean Sea, a unique alignment of ocean conditions allow strong winds to cool the ocean surface enough to cause mixing of the surface waters to great depths in a process known as deep convection. When this event ceases, the water column tends to restore its original configuration, with light waters above dense waters. We capture the onset of this process, known as restratification, in high-resolution observational detail together with measurements of ocean turbulence levels, collected with an autonomous ocean vehicle. The onset of restratification after convection takes place on a timescale of days to one week, and strongly influences turbulence variations within the surface waters. We conclude that this restratification has two main contributors, the near-surface stratification that forms due to the daily variation of surface buoyancy fluxes, and the lateral exchange of denser waters from the surroundings. With sufficient light and nutrients after convection, this near-surface stratification together with the absence of turbulent mixing allow phytoplankton to grow.

## 1. Introduction

Deep convection (DC) is a water mass transformation process that causes ocean surface waters to sink to great depths  $[O(10^3) \text{ m}]$  (e.g., Clarke & Gascard, 1983; MEDOC-Group, 1970), and this is mainly achieved by the interplay of the ocean circulation and atmospheric forcing, working together to weaken stratification in winter. DC primarily results in Deep Water formation, which plays a crucial role in regulating global climate by contributing to the thermohaline circulation through three-dimensional heat and salt transports (Marshall & Schott, 1999), and exchanging oxygen and carbon with the deeper ocean (Fröb et al., 2016; Ulses et al., 2021; Wolf et al., 2018). Few oceanic regions are prone to this process; we focus here on the Gulf of Lion, Northwestern Mediterranean Sea. After convection ceases, turbulent convective mixing gives way to a restratification process and, on a timescale of days to months, stratification is restored by surface buoyancy gain and lateral exchange between waters from the DC area and its surroundings (Marshall & Schott, 1999). Simultaneously, this newly-formed, near-surface stratification suppresses turbulence and vertical mixing, and together with a higher input of nutrients in the euphotic layer (d'Ortenzio et al., 2014; Kessouri et al., 2017) brought by convection, creates conditions conducive for phytoplankton growth the following spring (Frajka-Williams & Rhines, 2010; Lavigne





Figure 1. Chlorophyll-a (chl-a) maps from Moderate Resolution Imaging Spectroradiometer (MODIS) Aqua L3, 4 km, daily products in the Gulf of Lion averaged from (a) 1 to 31 January, (b) 14 and 17 February and (c) 20 and 27 February and the trajectory of the glider showing the transit (1) (yellow) and the deep convection center (11) (green) areas. While at (11), the glider transected until 28 February. Geostrophic currents (vectors) from Copernicus Marine Environment Monitoring Service (CMEMS, E.U. Copernicus Marine Service Information) are averaged between 8 and 28 February 2021. White contours represent chl-a = 0.25 mg m<sup>-3</sup>. Gaps in chl-a were linearly interpolated.

et al., 2013). However, stratification in the Mediterranean Sea is expected to increase due to increased heat and salt contents of its intermediate waters (Margirier et al., 2020) as a response to climate change (Adloff et al., 2015; Coma et al., 2009; Parras-Berrocal et al., 2022). In this context, and due to the significance of DC for global ocean circulation, it is essential to understand restratification and its interaction with ocean turbulence in DC regions.

Despite its relevance to upper ocean dynamics, vertically-resolved turbulence measurements in active DC regions have not, to our knowledge, been performed. This is likely due to the harsh conditions that make ship-based turbulent microstructure profiling difficult to impossible. For this reason many studies rely on models using turbulence parameterizations (André et al., 2005; Giordani et al., 2020; Pal & Chalamalla, 2020) to gain insight into the ocean's turbulent field. Here we overcome these limitations by performing turbulent microstructure measurements from an autonomous underwater glider (Fer et al., 2014; Palmer et al., 2015; Schultze et al., 2017) in the Gulf of Lion DC region. We show that strong variability is present in upper ocean turbulence that can be linked to the variable formation of diurnal stratification, and that relatively rapid time scales are found for the onset of this restratification in the near-surface. Finally, the shut down of turbulence by near-surface stratification provides ideal conditions for a nutrient-rich, newly-formed, shallow mixed layer to favor phytoplankton growth.

#### 2. Study Area and Methods

Our measurements took place in the Gulf of Lion from 8 to 28 February 2021. They were performed with a Teledyne Webb Research Slocum Electric G2 autonomous underwater glider *Comet*, equipped with custom Sea-Bird Electronics conductivity, temperature, and depth sensors (Seabird SBE41 CTD) measuring at 1 Hz, as well as various navigation and communications sensors. The glider sampled in a sawtooth pattern at an angle of approximately 26° to the horizontal. It collected a total of 1,452 profiles between  $\approx$ 2 and 200 m, with an average profile-to-profile horizontal distance of approximately 350 m. The glider surfaced roughly every 3 hr, for about 15 min, obtaining GPS positions and transferring decimated data sets to shore. Figure 1 shows the study area and the glider sampling locations. After the initial transit (①, Figure 1) to the DC region at around 42°N, 4.5°E, *Comet* was programmed to perform repeated transects around this location every 12 hr (①, DC area, Figure 1). Each transect has approximately 30 profiles and a horizontal distance of 10 km. We assessed mainly temporal and vertical variations within the study area. A second glider (*Dipsy*) sampling to a depth of 1,000 m was also deployed and followed a similar path to *Comet*. Some of these data are used, and further details are found in Text S2 of Supporting Information S1.

In addition to the glider-based sensors above, a Rockland Scientific MicroRider-1,000 for the measurement of turbulent microstructure was mounted on the glider *Comet*. The turbulence data reported here were obtained from

two air-foil shear probes (sampling at 512 Hz), a pressure transducer, electromagnetic current meter, two accelerometers, and an inclinometer measuring at 64 Hz. A time series of turbulent kinetic energy dissipation rate ( $\epsilon$ , [W kg<sup>-1</sup>]) is obtained from the microstructure shear measurements by accounting for the speed of flow past the sensors, due to the sensitivity of  $\epsilon$  on this parameter (Merckelbach & Carpenter, 2021; Merckelbach et al., 2019). The procedure used to arrive at  $\epsilon$  generally follows that of Schultze et al. (2017), and is described in Text S1 of Supporting Information S1. The MicroRider started sampling on 9 February, was turned off in an effort to conserve energy between 18 and 21 February, recorded only downcasts between 21 and 24 February, and malfunctioned toward the end of the experiment, resulting in some data gaps.

To evaluate the surface fluxes of momentum ( $\tau$ , [N m<sup>-2</sup>]) and buoyancy (B, [m<sup>2</sup> s<sup>-3</sup>]) we used the ERA5 reanalysis provided by the European Centre for Medium-Range Weather Forecasts (ECMWF, Hersbach et al., 2020). The surface heat flux (Q, [W m<sup>-2</sup>]) is calculated as the sum of the sensible and latent heat fluxes and solar radiation. The buoyancy flux is then found through the bulk formula in Grignon et al. (2010). As an additional tool to understand the role of surface fluxes in turbulence variability and stratification, we use the General Ocean Turbulence Model (GOTM, Umlauf & Burchard, 2005) to reproduce our observations in a one-dimensional framework (see Text S3 in Supporting Information S1 for a detailed description and results).

### 3. Results and Discussion

#### 3.1. The Deep Convection Area

The spatial extent of the Northwestern Mediterranean DC region is often quantified using satellite-derived surface chlorophyll-a (chl-a) levels (Herrmann et al., 2017; Houpert et al., 2016). The intense vertical mixing associated with convection results in reduced growth rates and a dilution of phytoplankton in the mixed layer, hence low surface chl-a visible from satellite imagery (the "blue hole" of Testor et al., 2018). We chose a surface chl-a threshold of 0.25 mg m<sup>-3</sup> to define the horizontal extent of the DC area based on previously reported values during periods of large buoyancy losses (Herrmann et al., 2017; Testor et al., 2018). In Figures 1a and 1b chl-a values of 0.25 mg m<sup>-3</sup> and smaller, contained by the 0.25 mg m<sup>-3</sup>-contour, are found to lie in the center of the DC region (~42°N, 4.5°E, Houpert et al., 2016), where the glider sampled.

Along the glider transit to the DC region (①, Figure 1), a gyre-scale structure can be seen: a steep, rising mixed layer base (Figures 2c and 2f) along the edge of the cyclonic circulation that corresponds to the strong Northern Current carrying surface fresh Atlantic Water (arrows in Figure 1). In this study, the mixed layer base is defined as the depth at which potential density changes 0.015 kg m<sup>-3</sup> relative to 10-m reference depth, the latter chosen such that any contribution from solar heating is avoided. The active mixing layer is the portion of the water column where turbulence is being sustained by surface fluxes. It is found to follow closely the depth at which potential density changes 0.0025 kg m<sup>-3</sup> relative to the surface (Figures 2c and 2d), as in Brainerd and Gregg (1995), which we use as a proxy. The observed gyre-scale circulation may result in upwards doming isopycnals that reduce stratification in the DC region, and allow for deep-reaching convection when strong winds blow and trigger large buoyancy losses at the surface (Marshall & Schott, 1999). This can partly be seen in the subsequent deepening of the mixed layer base, and a reduction in the measured stratification strength, as the glider entered the DC region around February 13 (see Figure 2c, and discussion below). We quantify the stratification strength by  $N^2 \equiv -(g/\rho_0) \partial \rho / \partial z$ , where  $\rho$  denotes the potential density with  $\rho_0$  a reference density, z the vertical coordinate (positive upwards), and g the gravitational acceleration.

The time the glider entered the DC region coincided with the period of strongest negative surface buoyancy fluxes during the measurement period (Figure 2b, with a cooling peaking at  $Q = -460 \text{ W m}^{-2}$ ), resulting in convective turbulence that reached deeper than the 200 m diving depth of the glider on February 14 and 15 (Figure 2d). Nearby observations from the second glider *Dipsy* show a mixed layer as deep as 300 m (see Text S2 in Supporting Information S1). A significant fraction of the deepening of the mixed layer base can be accounted for by the loss in buoyancy during the convection event, with the remaining presumed to be attributed to existing lateral gradients (GOTM simulation, Text S3 in Supporting Information S1). After the event, the daily-averaged surface buoyancy flux switches to predominantly positive (heating) conditions, but with considerable diurnal variability in the non-averaged time series of surface buoyancy flux, *B* (gray line, Figure 2b). In addition, the wind stress magnitude in Figure 2a shows that, with the exception of two events on February 22 and 27, winds remained calm





**Figure 2.** (a) Wind stress ( $\tau$ , black line) and wind magnitude at 10 m above surface (gray vectors), (b) hourly (gray) and daily averaged (blue, red lines) surface buoyancy fluxes (*B*) from ERA5, (c) squared buoyancy frequency ( $N^2$ ), (d) turbulent kinetic energy dissipation rates ( $\epsilon$ ), and (e) chlorophyll-a (chl-a, uncalibrated) from observations during transit (1) and in the Deep convection (DC) area (11). Please note the non-linearity of the vertical axis in panels (c) to (f). Orange vertical line marks the transition between transit and the DC area sampling. Black and gold lines in (c) and (d) shows the mixed layer and active mixing layer depths, respectively. (f) Schematic of the main features of interest. Triangles represent occurrences of diurnal warm layers.

following the change in sign of the averaged buoyancy flux, thus allowing for the restratification phase of the DC region (Houpert et al., 2016).

#### 3.2. Stratification and Turbulence Variability

The turbulence data show a strong variability that, due to the non-lagrangian nature of the glider measurements, corresponds to a mix of temporal and spatial components (Figure 2f). However, much of this variability can be understood through the interaction between the surface wind and buoyancy forcing, and the formation of diurnal warm layers (DWLs, Price et al., 1986). DWLs can be observed during periods of sufficiently weak winds and strong heating and are associated with the formation of a thin [O(1) m] layer of stratification (triangles, Figure 2c) that is dominated by temperature (Figure 3d). The presence of this stratification shows a rapid shutdown of the turbulence in the mixed layer below (Figures 2d and 3a) with high turbulence confined to the first few meters of



**Figure 3.** (a) Turbulence and (b) stratification of the diurnal warm layer on February 17 and the active mixing layer depth (gold) are shown as an example. Time evolution of (c) dissipation rates and (d) stratification, averaged between 2 and 10 m (red), and between 20 and 40 m (blue) from February 9, 10, 11, 13, 17, and 23. In (d), stratification is partitioned into its thermal (dot-dashed) and haline (dotted) contributions, highlighting the dominance of temperature in stratification in the upper 10 m. Vertical lines mark the change in sign of the surface heat flux, reaching its maximum at noon. Hatched areas correspond to the 95% confidence interval around the mean values. Black thick line in (c) is the linear regression fit estimate of the exponential decay of  $\epsilon$ , that is,  $\epsilon \sim e^{(-t/\gamma)}$ , within the 20–40 m layer.

the water column when the buoyancy flux is positive (Figures 3a and 3c). This shutdown implies that active mixing happens only in a very thin layer, although the mixed layer is as deep as 100 m (Figures 2c and 2d).

During the cooling phase the diurnal stratification can be mixed away, associated with a rapid increase in  $\epsilon$  to depths greater than 100 m (e.g., February 10, 14). The 1-D turbulence analysis shows that the variability of turbulence in the mixed layer is predominantly responding to changes in surface forcing (Text S3 in Supporting Information S1). To help understand the rapid increases in the mixed layer turbulence at depth, we look at a characteristic time scale, *T*, for the penetration of convective turbulence throughout the depth of the mixed layer. This time scale takes the form of  $T \sim H/w$ , where *H* is the mixed layer depth, and *w* a characteristic vertical velocity that is dependent on which terms represent the dominant balance in the convection. If buoyancy forcing is balanced by inertia we have  $w \sim (BH)^{1/3}$ , and T = O(1) hr, with  $B = 1 \times 10^{-7}$  m<sup>2</sup> s<sup>-3</sup> and H = 100 m (Marshall & Schott, 1999). On the other hand, if buoyancy forcing is balanced by Coriolis forces we find  $w \sim (B/f)^{1/2}$ , which leads to a similar result of T = O(1) hr (Fernando et al., 1991; Marshall & Schott, 1999). We therefore expect that the mixed layer turbulence responds rapidly over a time scale of ~1 hr to the mixing caused by convective turbulence following the loss of diurnal stratification, in agreement with the rapid increases in  $\epsilon$  in Figure 2d.

Figures 3c and 3d shows the time evolution of averaged dissipation and stratification in cases when a clear DWL could be identified (February 9, 10, 11, 13, 17, and 23) within two different depth ranges: between 2 and 10 m and between 20 and 40 m, the latter representing the "remnant" layer immediately below the bottom boundary of the DWL (Brainerd & Gregg, 1995). We excluded the remaining DWLs (February 19, 20, 24-26, 28) due to the scarcity (or absence) of dissipation rates. With the onset of the diurnal stratification, the DWL bottom boundary isolates the upper ocean from the remaining mixed layer, creating two distinct turbulence regimes (Brainerd & Gregg, 1993a). In the upper 10 m, mean  $\epsilon$  levels remain higher than  $10^{-7}$  W kg<sup>-1</sup>, slightly decreasing during heating conditions (red, Figure 3c), when stable stratification immediately forms (red, Figure 3d). In the deeper remnant layer, dissipation rates decrease one order of magnitude from the onset of the diurnal stratification until an hour after the peak of the heat flux (blue, Figure 3c). In other words, once isolated from the surface forcing by stratification, the averaged dissipation in the remnant layer shows a rapid decay with e-folding time scale of 140 min ( $R^2 = 0.92$ ), and may be associated with a weak restratification occurring within the remnant layer, where increases are seen from near the beginning of the heating period (Figures 3b and 3d). In stratified waters, turbulence is expected to exponentially decay once the background stratification is constant (Brainerd & Gregg, 1993b; Callaghan et al., 2014; Smyth et al., 1997). Our O(1) hr decay shows a rapid turbulence response to the near-surface stratification formation.







#### 3.3. Restratification: The Contribution of the Near-Surface

We have seen that upper ocean turbulence is highly variable in this region, and responds rapidly to varying nearsurface stratification over a time scale of roughly O(1) hr. The origins of this observed stratification are now explored and quantified using a buoyancy budget analysis. Recall that, once in the DC area, *Comet* performed repeated ~10-km transects ((II), Figure 1). Since these sections are much smaller compared to the DC area [O(100) km] and in its evolution [O(1) month], buoyancy variability should be dominated by temporal changes over spatial ones. The Rossby radius being around 10 km in the region, it is however not excluded that mesoscale processes contribute to this variability at smaller temporal scales.

Following the convection event of February 14, a rapid restratification takes place as the daily-mean heat gain becomes positive (Figure 2b). This restratification begins soon after February 16, when the buoyancy profile shows a completely mixed water column in the upper 200 m (①, Figure 4a; see also Figure S2 in Supporting Information S1). We use the buoyancy level of this fully mixed profile as a convenient reference ( $b_{mix}$ ) for the mixed state from which buoyancy anomalies  $b'(z, t) \equiv b(z, t) - b_{mix}$  are calculated. Figures 4a and 4b shows the evolution of the restratification phase through the buoyancy anomaly b'(z, t) as a function of depth and time. Using  $b_{mix}$ , we partition the restratification into positive and negative buoyancy anomalies that we associate with deep and shallow restratification components, respectively. Figures 4a and 4b shows that the buoyancy restratification proceeds over a time scale of T = O(1) week or less, and it is composed of both deep and shallow anomalies. We define the change in buoyancy content of a portion of the water column ( $z_b < z < z_t$ ) through

$$\tilde{b}(t) \equiv \int_{z_b}^{z_t} |b'(z,t)| \, dz,\tag{1}$$

where  $z_b$  and  $z_t$  are variable depth limits that yield the partial contribution of different layers to the total restratification. We evaluate the absolute buoyancy anomalies so that both deep and shallow restratification would have a positive contribution to the total.

It is helpful to look at various values for the top and bottom integration depths,  $z_t$  and  $z_b$ , respectively, and these are plotted in Figure 4c. Namely, the total buoyancy content anomaly ( $\tilde{b}_{tot}$ ) results from integration over the whole measured water column, the deep ( $\tilde{b}_{deep}$ ) and shallow ( $\tilde{b}_{shal}$ ) components come from integrating from the bottom



and top to the level  $z_{\text{mix}}$  where the mixed buoyancy is found, that is,  $b'(z_{\text{mix}}, t) = b_{\text{mix}}$  (red line, Figure 4b). The near-surface contribution we define as  $\tilde{b}_{ns}$ , the buoyancy anomaly content in the upper 20 m. This partitioning of the buoyancy content results in the sum of the deep and shallow contributions giving the total, that is,  $\tilde{b}_{\text{tot}} = \tilde{b}_{\text{deep}} + \tilde{b}_{\text{shal}}$ . Additionally, we can include the contribution of the surface fluxes through

$$\tilde{B}(t) \equiv \int_{t_0}^t B(t') dt',$$
(2)

where  $t_0$  represents the beginning of the restratification period on 16 February (Figure 4c). Figure 4c shows that from February 16 until the formation of near-surface restratification on February 19, the integrated surface buoyancy flux is negative ( $\tilde{B}$ , Figure 4c) and restratification is mainly through lateral advection, with inputs of buoyancy over the 0–200 m layer (Figures 4b and 4c). By the end of the restratification period shown, the nearsurface (upper 20 m) component accounts for around 19% of the total stratification, with 59% of that value due to the surface buoyancy flux ( $\tilde{B}$ ). Interestingly, more than half (60%) of the total buoyancy anomaly content of a 200 m-deep water column is found roughly in the first 100 m, showing the significance of shallow water processes and their interplay with atmospheric forcing.

A remarkable feature of the evolution of the buoyancy content in Figure 4 are the relatively rapid time scales that are present in restratifying the water column after the February 14-15 convection event, with both deep and shallow increases in buoyancy occurring over days to O(1) week. A time scale for the baroclinic mesoscale eddydriven restratification of convection regions has been proposed by Jones and Marshall (1997) as  $T_{\text{restrat}} \approx 56 r/$  $(\Delta bh)^{1/2}$ , with r a radial length scale for a patch of dense fluid with buoyancy difference  $\Delta b$  from the ambient fluid, and a depth h. Taking h = 300 m,  $\Delta b = 3 \cdot 10^{-4} \text{ m s}^{-2}$  representative of the deep buoyancy anomaly, and  $r \approx 50$  km (see Figure 1), this results in  $T_{\text{restrat}} = O(100)$  days. Although this time scale is representative of the full restratification of the region through mesoscale eddies, there are also found to be faster restratification mechanisms operating (e.g., Clemént et al., 2023; Houpert et al., 2016; Testor & Gascard, 2006; Testor et al., 2018), such as the advection of stratified waters by wind-driven Ekman (Bosse et al., 2021) and submesoscale (Boccaletti et al., 2006; Clemént et al., 2023; Testor & Gascard, 2006) processes, and through lateral changes in temperature and salinity being converted to vertical changes (Brainerd & Gregg, 1993b; Caldwell et al., 1997). Using a baroclinic instability model, Testor and Gascard (2006) found a timescale of ~6 days for the restratification. Couvelard et al. (2015) showed large differences in restratification time scales depending on a model's spatial resolution, hinting at the importance of smaller scale processes; with a 2 km resolution, restratification could occur in less than 10 days. In the buoyancy analysis of Figure 4 we point to the importance of lateral advective processes in both the shallow and near-surface restratification observed. Only roughly 60% of the near-surface buoyancy gain  $(b_{ns})$  can be accounted for by surface fluxes, with the remaining 40% accounted for presumably through lateral advective processes (see additionally Figure S3 in Supporting Information S1). This fraction becomes much larger ( $\approx 80\%$ ) for the shallow buoyancy gain,  $\tilde{b}_{shal}$ , and has important links to the growth of phytoplankton from restratification processes.

#### 3.4. Implications for Phytoplankton Growth

In a classical framework, phytoplankton is expected to bloom when the mixed layer depth shoals beyond a critical value, due to changes in the surface heat flux in spring from cooling to heating (Franks, 2015; Sverdrup, 1953). However, high phytoplankton concentrations may appear before spring, related to a higher input of nutrients from convection events (Figure 1, Kessouri et al., 2018; Severin et al., 2014), the decrease in turbulence levels in the mixed layer (Franks, 2015; Taylor & Ferrari, 2011) and also the feedbacks between phytoplankton and its predators (Behrenfeld & Boss, 2014). The rapid shoaling of the mixed layer and the restratification of the near-surface that we observe following the convection event (Figures 2 and 4) traps the turbulence in a very thin (<10 m) layer, preventing mixing and allowing phytoplankton to grow (e.g., Kessouri et al., 2018; Ulses et al., 2016). Figure 2e shows the variations in chl-a concentration throughout our time series. Since the sensor was not calibrated prior to deployment, the chl-a estimates should be interpreted with caution, and primarily in qualitative terms. Quenching effects were corrected following Thomalla et al. (2018). One can readily observe higher values of chl-a associated with stratified conditions and lower chl-a levels associated with a mixed state. In particular, it can be seen that increases in chl-a occur when near-surface stratified layers have formed during

restratification (e.g., February 19, 25, 26 in Figures 2c and 2e). These layers have turbulence that is also confined only to the near-surface layer, and are unable to mix plankton throughout the generally deeper mixed layer. This situation may be contrasted to the two wind events on February 22 and 27 where chl-a concentrations are found to be homogeneous throughout the mixed layer.

The formation of near-surface stratification largely shuts down turbulence throughout the water column, leading to the growth of phytoplankton within the euphotic zone (i.e., not confined solely to the stratified region near the surface) over a time scale of days. This is a similar mechanism to that suggested by Taylor and Ferrari (2011), who connected the convective forcing associated with wintertime cooling to turbulent mixing, and hypothesized that phytoplankton blooms can be triggered when the net heat flux switches from cooling to heating. We find this switch to occur through the rapid formation of near-surface stratification and its control on upper ocean turbulence. Through such a mechanism, phytoplankton can be trapped in the deeper stratified ocean, as is seen in Figure 2e below the mixed layer base. This could represent a carbon export mechanism associated with restratification (e.g., Lacour et al., 2019).

## 4. Summary and Conclusions

Turbulence measurements were collected in the DC region of the Gulf of Lion using an autonomous ocean glider equipped with a turbulent microstructure package. These unique observations provide a detailed characterization of the mixed layer during a 300-m deep convective event, followed by the subsequent restratification of the DC region. The variability in upper ocean turbulence over timescales of hours to days is strongly linked to both changes in surface forcing, and near-surface restratification processes. The near-surface stratification confines the turbulence in the upper meters of the water column, and isolates deeper layers from surface-driven turbulence. Together with the relatively rapid growth and decay timescales of the turbulence [O(1) hr], these processes are capable of producing the strong variability in upper ocean turbulence in the convection region. Restratification of the upper 200 m of the water column occurs over a timescale of O(1) week through both the diurnal variability in surface buoyancy fluxes and through lateral advective processes. The formation of near-surface stratification and its effects on the turbulence distribution are also found to allow for an effective growth of phytoplankton. Thus we find that complex, and rapidly varying, near-surface processes are crucial to understand restratification and its biophysical feedbacks in the DC region.

## **Data Availability Statement**

The glider data used in this study is available at Miracca-Lage (2023). The wind and surface fluxes data from ERA5 reanalysis provided by the European Centre for Medium-Range Weather Forecasts (ECMWF) are accessible through Hersbach et al. (2022). The chlorophyll-a data from MODIS-Aqua and the altimeric products are accessible through NASA Goddard Space Flight Center, Ocean Ecology Laboratory, Ocean Biology Processing Group (2022) and E.U. Copernicus Marine Service Information, CMEMS (2022), respectively.

#### Acknowledgments

Funding for this work has been provided through the Helmholtz Association PoF-IV program, as well as being a contribution to project T2 (Ocean Surface Layer Energetics) of the Collaborative Research Center TRR 181 "Energy transfers in Atmosphere and Ocean" funded by the Deutsche Forschungsgemeinschaft (DFG, German Research Foundation) -Projektnummer 274762653. The glider data analysis received support from GROOM-RI funded by the European Union's Horizon 2020 research and innovation program under Grant agreement no. 951842. Open Access funding enabled and organized by Projekt DEAL.

#### References

- Adloff, F., Somot, S., Sevault, F., Jordà, G., Aznar, R., Déqué, M., et al. (2015). Mediterranean Sea response to climate change in an ensemble of twenty first century scenarios. *Climate Dynamics*, 45(9–10), 2775–2802. https://doi.org/10.1007/s00382-015-2507-3
- André, G., Garreau, P., Garnier, V., & Fraunié, P. (2005). Modelled variability of the sea surface circulation in the North-western Mediterranean Sea and in the Gulf of Lions. Ocean Dynamics, 55(3–4), 294–308. https://doi.org/10.1007/s10236-005-0013-6
  - Behrenfeld, M. J., & Boss, E. S. (2014). Resurrecting the ecological underpinnings of ocean plankton blooms. *Annual Review of Marine Science*, 6(1), 167–194. https://doi.org/10.1146/annurev-marine-052913-021325
     Boccaletti, G., Ferrari, R., & Fox-Kemper, B. (2006). Mixed layer instabilities and restratification. *Journal of Physical Oceanography*, 37(9),
  - Boccaletti, G., Ferrari, K., & Fox-Kemper, B. (2006). Mixed layer instabilities and restratification. *Journal of Physical Oceanography*, 57(9), 2228–2250. https://doi.org/10.1175/JPO3101.1
  - Bosse, A., Testor, P., Damien, P., Estournel, C., Marsaleix, P., Mortier, L., et al. (2021). Wind-forced submesoscale symmetric instability around deep convection in the northwestern Mediterranean Sea. *Fluids*, *6*(3), 123. https://doi.org/10.3390/fluids6030123
  - Brainerd, K., & Gregg, M. (1993a). Diurnal restratification and turbulence in the oceanic surface mixed layer: 1. Observations. Journal of Geophysical Research, 98(C12), 22645–22656. https://doi.org/10.1029/93JC02297
  - Brainerd, K., & Gregg, M. (1993b). Diurnal restratification and turbulence in the oceanic surface mixed layer. II: Modeling. *Journal of Geophysical Research*, 98(C12), 22657–22664. https://doi.org/10.1029/93JC02298
  - Brainerd, K., & Gregg, M. (1995). Surface mixed and mixing layer depths. *Deep Sea Research Part I: Oceanographic Research Papers*, 42(9), 1521–1543. https://doi.org/10.1016/0967-0637(95)00068-H
  - Caldwell, D., Lien, R.-C., Moum, J. N., & Gregg, M. C. (1997). Turbulence decay and restratification in the equatorial ocean surface layer following nighttime convection. *Journal of Physical Oceanography*, 27(6), 1120–1132. https://doi.org/10.1175/1520-0485(1997)027(1120: TDARIT)2.0.CO;2

- Callaghan, A. H., Ward, B., & Vialard, J. (2014). Influence of surface forcing on near-surface and mixing layer turbulence in the tropical Indian Ocean. *Deep Sea Research Part I: Oceanographic Research Papers*, 94, 107–123. https://doi.org/10.1016/j.dsr.2014.08.009
- Clarke, R. A., & Gascard, J.-C. (1983). The formation of Labrador Sea water. Part I: Large-scale processes. Journal of Physical Oceanography, 13(10), 1764–1778. https://doi.org/10.1175/1520-0485(1983)013(1764:TFOLSW)2.0.CO;2
- Clemént, L., Frajka-Williams, E., von Oppeln-Bronikowski, N., Goszczko, I., & de Young, B. (2023). Cessation of Labrador Sea convection triggered by distinct fresh and warm (sub) mesoscale flows. *Journal of Physical Oceanography*, 53(8), 1959–1977. https://doi.org/10.1175/ JPO-D-22-0178.1
- Coma, R., Ribes, M., Serrano, E., Jiménez, E., Salat, J., & Pascual, J. (2009). Global warming-enhanced stratification and mass mortality events in the Mediterranean. *Proceedings of the National Academy of Sciences*, *106*(15), 6176–6781. https://doi.org/10.1073/pnas.0805801106
- Couvelard, X., Dumas, F., Garnier, V., Ponte, A. L., Talandier, C., & Treguier, A. M. (2015). Mixed layer formation and restratification in presence of mesoscale and submesoscale turbulence. *Ocean Modelling*, 96, 243–253. https://doi.org/10.1016/j.ocemod.2015.10.004
- d'Ortenzio, F., Lavigne, H., Besson, F., Claustre, H., Coppola, L., Garcia, N., et al. (2014). Observing mixed layer depth, nitrate and chlorophyll concentrations in the northwestern Mediterranean: A combined satellite and NO<sub>3</sub> profiling floats experiment. *Geophysical Research Letters*, 41(18), 6443–6451. https://doi.org/10.1002/2014GL061020
- E.U. Copernicus Marine Service Information, CMEMS. (2022). Global Ocean gridded L4 sea surface heights and derived variables NRT. Copernicus Marine Service. https://doi.org/10.48670/moi-00149
- Fer, I., Petersen, A. K., & Ullgren, J. E. (2014). Microstructure measurements from an underwater glider in the turbulent Faroe Bank Channel overflow. Journal of Atmospheric and Oceanic Technology, 31(5), 1128–1150. https://doi.org/10.1175/JTECH-D-13-00221.1
- Fernando, H. J., Chen, R.-R., & Boyer, D. L. (1991). Effects of rotation on convective turbulence. Journal of Fluid Mechanics, 228, 513–547. https://doi.org/10.1017/S002211209100280X
- Frajka-Williams, E., & Rhines, P. B. (2010). Physical controls and interannual variability of the Labrador Sea spring phytoplankton bloom in distinct regions. Deep Sea Research Part I: Oceanographic Research Papers, 57(4), 541–552. https://doi.org/10.1016/j.dsr.2010.01.003
- Franks, P. J. S. (2015). Has Sverdrup's critical depth hypothesis been tested? Mixed layers vs. turbulent layers. ICES Journal of Marine Science, 72(6), 1897–1907. https://doi.org/10.1093/icesjms/fsu175
- Fröb, F., Olsen, A., Våge, K., Moore, G., Yashayaev, I., Jeansson, E., & Rajasakaren, B. (2016). Irminger Sea deep convection injects oxygen and anthropogenic carbon to the ocean interior. *Nature Communications*, 7(1), 13244. https://doi.org/10.1038/ncomms13244
- Giordani, H., Bourdallé-Badie, R., & Madec, G. (2020). An eddy-diffusivity mass-flux parameterization for modeling oceanic convection. Journal of Advances in Modeling Earth Systems, 12(9), e2020MS002078. https://doi.org/10.1029/2020ms002078
- Grignon, L., Smeed, D. A., Bryden, H. L., & Schroeder, K. (2010). Importance of the variability of hydrographic preconditioning for deep convection in the Gulf of Lion, NW Mediterranean. Ocean Science, 6(2), 573–586. https://doi.org/10.5194/os-6-573-2010
- Herrmann, M., Auger, P.-A., Ulses, C., & Estournel, C. (2017). Long-term monitoring of ocean deep convection using multisensors altimetry and ocean color satellite data. *Journal of Geophysical Research: Oceans*, 122(2), 1457–1475. https://doi.org/10.1002/2016JC011833
- Hersbach, H., Bell, B., Berrisford, P., Biavati, G., Horányi, A., Muñoz Sabater, J., et al. (2022). ERA5 hourly data on single levels from 1940 to present. Copernicus Climate Change Service (C3S) Climate Data Store (CDS). https://doi.org/10.24381/cds.adbb2d47
- Hersbach, H., Bell, B., Berrisford, P., Hirahara, S., Horányi, A., Muñoz-Sabater, J., et al. (2020). The ERA5 global reanalysis. *Quarterly Journal of the Royal Meteorological Society*, 146(730), 1999–2049. https://doi.org/10.1002/qj.3803
- Houpert, L., Durrieu de Madron, X., Testor, P., Bosse, A., d'Ortenzio, F., Bouin, M.-N., et al. (2016). Observations of open-ocean deep convection in the northwestern Mediterranean S ea: Seasonal and interannual variability of mixing and deep water masses for the 2007–2013 period. *Journal of Geophysical Research: Oceans*, 121(11), 8139–8171. https://doi.org/10.1002/2016JC011857
- Jones, H., & Marshall, J. (1997). Restratification after deep convection. Journal of Physical Oceanography, 27(10), 2276–2287. https://doi.org/10. 1175/1520-0485(1997)027(2276:RADC)2.0.CO;2
- Kessouri, F., Ulses, C., Estournel, C., Marsaleix, P., d'Ortenzio, F., Severin, T., et al. (2018). Vertical mixing effects on phytoplankton dynamics and organic carbon export in the western Mediterranean Sea. *Journal of Geophysical Research: Oceans*, 123(3), 1647–1669. https://doi.org/10. 1002/2016JC012669
- Kessouri, F., Ulses, C., Estournel, C., Marsaleix, P., Séverin, T., Pujo-Pay, M., et al. (2017). Nitrogen and phosphorus budgets in the Northwestern Mediterranean deep convection region. *Journal of Geophysical Research: Oceans*, 122(12), 9429–9454. https://doi.org/10.1002/2016jc012665
- Lacour, L., Briggs, N., Claustre, H., Ardyna, M., & Dall'Olmo, G. (2019). The intraseasonal dynamics of the mixed layer pump in the subpolar North Atlantic Ocean: A biogeochemical-Argo float approach. *Global Biogeochemical Cycles*, 33(3), 266–281. https://doi.org/10.1029/ 2018GB005997
- Lavigne, H., d'Ortenzio, F., Migon, C., Claustre, H., Testor, P., d'Alcalà, M. R., et al. (2013). Enhancing the comprehension of mixed layer depth control on the Mediterranean phytoplankton phenology. *Journal of Geophysical Research: Oceans*, 118(7), 3416–3430. https://doi.org/10. 1002/jgrc.20251
- Margirier, F., Testor, P., Heslop, E., Mallil, K., Bosse, A., Houpert, L., et al. (2020). Abrupt warming and salinification of intermediate waters interplays with decline of deep convection in the Northwestern Mediterranean Sea. *Nature Scientific Reports*, 10, 20923. https://doi.org/10. 1038/s41598-020-77859-5
- Marshall, J., & Schott, F. (1999). Open-ocean convection: Observations, theory, and models. *Reviews of Geophysics*, 37(1), 1–64. https://doi.org/ 10.1029/98RG02739
- Medoc-Group, T. (1970). Observation of formation of deep water in the Mediterranean Sea, 1969. Nature, 227(5262), 1037–1040. https://doi.org/ 10.1038/2271037a0
- Merckelbach, L., Berger, A., Krahmann, G., Dengler, M., & Carpenter, J. R. (2019). A dynamic flight model for Slocum gliders and implications for turbulence microstructure measurements. *Journal of Atmospheric and Oceanic Technology*, 36(2), 281–296. https://doi.org/10.1175/ JTECH-D-18-0168.1
- Merckelbach, L., & Carpenter, J. R. (2021). Ocean glider flight in the presence of surface waves. Journal of Atmospheric and Oceanic Technology, 38(7), 1265–1275. https://doi.org/10.1175/jtech-d-20-0206.1
- Miracca-Lage, M. (2023). Processed data for a Slocum ocean glider campaign in the Gulf of lion, Mediterranean Sea [Dataset]. Zenodo. https:// doi.org/10.5281/zenodo.10133152
- NASA Goddard Space Flight Center, Ocean Ecology Laboratory, Ocean Biology Processing Group. (2022). Moderate-resolution imaging spectroradiometer (MODIS) Aqua global ocean gridded daily level-3 ocean color data, chlorophyll concentration, 4 km resolution. NASA OB. DAAC. https://doi.org/10.5067/AQUA/MODIS/L3M/CHL/2022
- Pal, A., & Chalamalla, V. K. (2020). Evolution of plumes and turbulent dynamics in deep-ocean convection. *Journal of Fluid Mechanics*, 889(A35), 1–26. https://doi.org/10.1017/jfm.2020.94



- Palmer, M. R., Stephenson, G. R., Inall, M. E., Balfour, C., Düsterhus, A., & Green, J. A. M. (2015). Turbulence and mixing by internal waves in the Celtic Sea determined from ocean glider microstructure measurements. *Journal of Marine Systems*, 144, 57–69. https://doi.org/10.1016/j. jmarsys.2014.11.005
- Parras-Berrocal, I. M., Vázquez, R., Cabos, W., Sein, D. V., Álvarez, O., Bruno, M., & Izquierdo, A. (2022). Surface and intermediate water changes triggering the future collapse of deep water formation in the North Western Mediterranean. *Geophysical Research Letters*, 49(4), e2021GL095404. https://doi.org/10.1029/2021GL095404
- Price, J. F., Weller, R. A., & Pinkel, R. (1986). Diurnal cycling: Observations and models of the upper ocean response to diurnal heating, cooling, and wind mixing. *Journal of Geophysical Research*, 91(C7), 8411–8427. https://doi.org/10.1029/JC091iC07p08411
- Schultze, L. K. P., Merckelbach, L. M., & Carpenter, J. R. (2017). Turbulence and mixing in a shallow shelf sea from underwater gliders. *Journal of Geophysical Research: Oceans*, 122(11), 9092–9109. https://doi.org/10.1002/2017JC012872
- Severin, T., Conan, P., De Madron, X. D., Houpert, L., Oliver, M. J., Oriol, L., et al. (2014). Impact of open-ocean convection on nutrients, phytoplankton biomass and activity. Deep Sea Research Part I: Oceanographic Research Papers, 94, 62–71. https://doi.org/10.1016/j.dsr. 2014.07.015
- Smyth, W. D., Zavialov, P. O., & Moum, J. N. (1997). Decay of turbulence in the upper ocean following sudden isolation from surface forcing. Journal of Physical Oceanography, 27(5), 810–822. https://doi.org/10.1175/1520-0485(1997)027(0810:DOTITU)2.0.CO;2
- Sverdrup, H. U. (1953). On conditions for the vernal blooming of phytoplankton. Journal du Conseil International pour l'Exploration de la Mer, 18(3), 287–295. https://doi.org/10.1093/icesjms/18.3.287
- Taylor, J. R., & Ferrari, R. (2011). Shutdown of turbulent convection as a new criterion for the onset of spring phytoplankton blooms. Limnology & Oceanography, 56(6), 2293–2307. https://doi.org/10.4319/lo.2011.56.6.2293
- Testor, P., Bosse, A., Houpert, L., Margirier, F., Mortier, L., Legoff, H., et al. (2018). Multiscale observations of deep convection in the northwestern Mediterranean Sea during winter 2012–2013 using multiple platforms. *Journal of Geophysical Research: Oceans*, 123(3), 1745–1776. https://doi.org/10.1002/2016JC012671
- Testor, P., & Gascard, J.-C. (2006). Post-convection spreading phase in the Northwestern Mediterranean Sea. Deep Sea Research Part I: Oceanographic Research Papers, 53(5), 869–893. https://doi.org/10.1016/j.dsr.2006.02.004
- Thomalla, S. J., Moutier, W., Ryan-Keogh, T. J., Gregor, L., & Schütt, J. (2018). An optimized method for correcting fluorescence quenching using optical backscattering on autonomous platforms. *Limnology and Oceanography: Methods*, 16(2), 132–144. https://doi.org/10.1002/lom3. 10234
- Ulses, C., Auger, P.-A., Soetaert, K., Marsaleix, P., Diaz, F., Coppola, L., et al. (2016). Budget of organic carbon in the North-Western Mediterranean open sea over the period 2004–2008 using 3-D coupled physical-biogeochemical modeling. *Journal of Geophysical Research: Oceans*, 121(9), 7026–7055. https://doi.org/10.1002/2016JC011818
- Ulses, C., Estournel, C., Fourrier, M., Coppola, L., Kessouri, F., Lefèvre, D., & Marsaleix, P. (2021). Oxygen budget of the north-western Mediterranean deep-convection region. *Biogeosciences*, 18(3), 937–960. https://doi.org/10.5194/bg-18-937-2021
- Umlauf, L., & Burchard, H. (2005). Second-order turbulence closure models for geophysical boundary layers. A review of recent work. Continental Shelf Research, 25(7–8), 795–827. https://doi.org/10.1016/j.csr.2004.08.004
- Wolf, M. K., Hamme, R. C., Gilbert, D., Yashayaev, I., & Thierry, V. (2018). Oxygen saturation surrounding deep water formation events in the Labrador Sea from Argo-O<sub>2</sub> data. *Global Biogeochemical Cycles*, 32(4), 635–653. https://doi.org/10.1002/2017GB005829

## **References From the Supporting Information**

- Kondo, J. (1975). Air-sea bulk transfer coefficients in diabatic conditions. Boundary-Layer Meteorology, 9(1), 91–112. https://doi.org/10.1007/ BF00232256
- Lueck, R. (2013). Calculating the rate of dissipation of turbulent kinetic energy (Technical Report Nos. TN-028) (p. 18). Rockland Scientific International Inc. Retrieved from http://rocklandscientic.com/?wpdmdl51034
- Wolk, F., Yamazaki, H., Seuront, L., & Lueck, R. G. (2002). A new free-fall profiler for measuring biophysical microstructure. Journal of Atmospheric and Oceanic Technology, 19(5), 780–793. https://doi.org/10.1175/1520-0426(2002)019h0780:ANFFPFi2.0.CO;2